General Practitioners perspectives on infant telomere length screening after a pregnancy complication: a qualitative analysis.

Carolyn J. Puglisi¹, Joshua McDonough², Tina Bianco-Miotto^{1,3,1}, Jessica A. Grieger^{3,4,a,*,}

¹School of Agriculture, Food and Wine, University of Adelaide, Adelaide 5005, Australia

²School of Public Health, University of Adelaide, Adelaide 5005, Australia

³Robinson Research Institute, University of Adelaide, Adelaide 5005, Australia

⁴Adelaide Medical School, University of Adelaide, Adelaide 5005, Australia

*Corresponding author: Adelaide Medical School and the Robinson Research Institute, The University of Adelaide, Adelaide 5005, Australia. E-mail: jessica. grieger@adelaide.edu.au

°0000-0003-1515-948X

Background: Pregnancy complications can impact the mother and child's health in the short and longterm resulting in an increased risk of chronic disease later in life. Telomere length is a biomarker of future cardiometabolic diseases and may offer a novel way of identifying offspring most at risk for future chronic diseases.

Objective(s): To qualitatively explore General Practitioners' (GPs) perspectives on the feasibility and uptake for recommending a telomere screening test in children who were born after a pregnancy complication.

Methods: Twelve semi-structured interviews were conducted with GPs within metropolitan Adelaide, South Australia. Interviews were audio recorded, transcribed verbatim, and analysed for codes and themes.

Results: Two themes were generated: ethical considerations and practical considerations. Ethically, the GP participants discussed barriers including consenting on behalf of a child, parental guilt, and the impact of health insurance, whereas viewing it for health promotion was a facilitator. For practical considerations, barriers included the difficulty in identifying people eligible for screening, maintaining medical communication between service providers, and time and financial constraints, whereas linking screening for telomere length with existing screening would facilitate uptake.

Conclusions: GPs were generally supportive of potential telomere screening in infants, particularly via a saliva test that could be embedded in current antenatal care. However, several challenges, such as lack of knowledge, ethical considerations, and time and financial constraints, need to be overcome before such a test could be implemented into practice.

Lay summary

Around one fifth of women experience a pregnancy complication that places their infants at higher risk for a range of chronic diseases in later life. Although not all infants will have adverse health outcomes, it is important to identify offspring early in life who may be at higher risk. Telomere length is a biomarker of future chronic disease that can be obtained from blood or saliva. Whether telomere length might be useful as a screening tool in newborns born from a pregnancy complication has not been investigated.

This study qualitatively explores the perspectives of 12 general practitioners in Adelaide, Australia, on the feasibility and uptake of telomere screening in children born after a pregnancy complication. Overall, general practitioners were generally supportive of potential telomere screening in infants, particularly via a saliva test that could be embedded in current antenatal care. However, several challenges, such as lack of knowledge, ethical considerations, and time and financial constraints, need to be overcome before such a test could be implemented into practice.

Study findings contribute to the limited knowledge assessing follow-up of screening after pregnancy within Australia and internationally and provide novel findings on a potential new screening tool that could be considered early in life.

Key words: Australia, GPs, infant screening, pregnancy complication, telomere length

Introduction

In Australia, around 25% of all pregnancies are complicated by major pregnancy complications such as gestational diabetes mellitus, preeclampsia, pre-term birth, and intrauterine growth restriction,¹ contributing to increased risk for several cardiometabolic diseases in offspring.^{2,3} Offspring born after a pregnancy complication may be born large or small for gestational age,^{4,5} have increased risk for perinatal morbidity and mortality,⁶ but they are also at increased risk for obesity and insulin resistance,⁷ type 2 diabetes,⁸ and hypertension⁹ in childhood or later life. While not all infants born from a pregnancy complication will have adverse health outcomes, there is a clear need to identify offspring early in life who may be at risk for potential adverse health outcomes in the longer term.

The identification of those most at risk is complex; disease may not manifest until adulthood, but the feasibility

[©] The Author(s) 2023. Published by Oxford University Press.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Key messages

- General Practitioners are supportive to include a newborn telomere screening test into postnatal care.
- Barriers include consent, parental guilt, and the impact of health insurance.
- Education programmes and the relevance of follow-up screening in children are needed.

of monitoring every child would impose a large strain on healthcare resources. Birthweight is a risk factor for future disease, with a 14% increased risk of cardiovascular disease in adulthood being reported for low birthweight and an 8% increased risk for high birthweight.¹⁰ However, birthweight has not been shown to produce policy changing evidence for its use as a tool for predicting future disease risk.¹¹ A screening tool to identify children who are most at risk would improve long-term health outcomes by identifying individuals that should be monitored for the disease.

Measurement of telomere length offers a novel way of identifying offspring who may be at greater risk of chronic disease. Telomeres are repeating sequences of nucleotides found at the ends of chromosomes, and their main function is to maintain chromosomal integrity.¹² Several studies in adults demonstrate shorter telomere length is associated with diabetes, cardiovascular disease, and cardiovascular mortality.^{13,14} Pregnancy complications such as gestational diabetes,¹⁵ preeclampsia,¹⁶ intrauterine growth restriction,¹⁷ and even maternal metabolic syndrome,¹⁸ are associated with shorter telomere length in offspring. Thus, the measurement of telomere length, a biomarker of chronic diseases, has the potential to identify children with a higher risk of future disease.

The role of the General Practitioner (GP) in providing care for all people, including postnatal care is vital in optimising outcomes for families. GPs are accessible and affordable for the majority of the population.¹⁹ In lieu of a national approach to follow up after a pregnancy complication, screening for telomere length in general practice may be effective, as more than 90% of Australians visit a general practitioner at least once a year.¹⁹ A previous study amongst health professionals in New South Wales (Australia), highlighted barriers towards child developmental surveillance programmes and screening tools, including time, tool awareness, knowledge, and access to referral pathways.²⁰ A systematic review on the acceptability of a range of childhood screening interventions found parents had doubts in the effectiveness of the test or in the accuracy of results (n = 6 studies), however, when screening might occur in venues that required minimal effort from the parent, the acceptability was increased $(n = 3 \text{ studies})^{21}$ Whether telomere length might be useful as a screening tool in newborns born from a pregnancy complication has not been investigated. This study aims to qualitatively explore GPs perspectives on the feasibility and uptake for recommending a telomere screening test in children who were born after a pregnancy complication.

Methods

Design

A qualitative design using semi-structured interviews was employed to elicit the views of general practitioners on the usefulness and feasibility of a blood test to measure telomere length in children born of a complicated pregnancy to determine the risk of future chronic disease.

Participants and recruitment

GPs within South Australia were purposively recruited; (i) through practice managers by email communication and (ii) through collaboration with a not-for-profit organisation, specialising in GP and GP Obstetrician education and support. Our research was presented at education sessions and expressions of interest registered. Eligibility criteria included GPs who were currently practicing in South Australia and had practiced for at least one year in South Australia and could communicate fluently in English. Recruitment took place between November 2021 and March 2022. Initially, 19 general practices were contacted by phone, and information packages were sent by email to practice managers. Emails were followed up with phone communication to ensure information was received and distributed to GPs. Recruitment was slow in the first 2-3 months, however, this was during a recent surge of the COVID-19 pandemic in Australia. Only four GPs were recruited through practice managers. We then focussed on snowballing recruitment via GP Partners, a notfor-profit organisation specialising in shared care education and training of GPs and GP-obstetricians. Two PowerPoint presentations at two GP Partners' education sessions were delivered to approximately 50 GPs to explain the study and establish a rapport with potential interviewees. GPs were offered a \$50 gift voucher to participate. In total, 14 GPs expressed interest [10 who were affiliates of GP Partners and 4 from smaller GP clinics], and all met study eligibility. Two GPs declined an interview due to personal reasons or time restrictions. All GPs consented to have their interview audiorecorded, except one, for personal reasons, but a thorough, verbatim, written transcript was collected and included in the analysis. All interviewees provided verbal consent. Ethical approval was granted by The University of Adelaide Human Research Ethics Committee [H-2021-110, date approved 2021 October 26].

Data collection

This study was guided by the Consolidated criteria for reporting qualitative studies (COREQ),²² to enhance research validity (purposeful sampling, audit trail), rigour (data saturation, ethics approval), credibility (member checking), and generality (inclusion criteria). Interviews were semi-structured based on the study objectives. Each GP described their knowledge regarding pregnancy complications and any relevant follow-up guidelines/recommendations; attitudes regarding the potential implementation of a screening test in children; facilitators and barriers regarding potential usage and uptake. The interviews utilised a reflective approach²³ allowing flexibility for the researcher to either elaborate or clarify certain responses without a pre-established assumption. The interview was facilitated by an interview schedule

with open-ended questions designed to allow participants to respond, while also elaborating on areas they considered to be important. Several guiding questions were created to ensure that the interview generated relevant data. Basic demographic data, including age, gender, years of practice in medicine/GP, and location (metropolitan/rural) was collected from each participant. Prior to commencement, the questions were pilot tested on two researchers and two GPs, with slight modifications made to improve clarity. The semi-structured telephone interviews were conducted by one researcher (CP). following training by an experienced qualitative researcher (JM). CP is a research scientist with experience in acute care (intensive care and high dependency) hospital ward work (both surgical and medical), and interview conduct. JM is a PhD graduate with expertise in qualitative data analysis, and qualitative research in the GP setting. Third author, TBM, is a senior research scientist with expertise in epigenetics (including telomere length). The last author, JAG, is a midcareer research scientist with experience in qualitative data collection and analysis, and telomere length biology. A single interviewer allowed for continual and simultaneous data collection, cross-checking of themes, and confirmation of data saturation. No interviews were repeated. GPs were given the opportunity to review the transcripts but provided no further comments.

Data analysis

De-identified audio recordings were reviewed by CP [a trained research fellow] for accuracy and were transcribed using a professional transcription service (Pacific Transcription). Transcripts were thematically analysed and coded using NVivo version 12 Plus (Windows) 2018 QSR International Pty Ltd Software. Using the Braun and Clarke method,²⁴ familiarising the transcript is the first step in data analysis, followed by the identification of elements of interest in the data. These initial codes were then linked together to create themes. Final themes were reviewed to ensure they were reflective of the original transcript and the research question. JM coded all transcripts, CP independently coded a third of the transcripts, which were then reviewed and compared with coding by JM to ensure consistency and reliability. Discussions between investigators were conducted three times to gain a consensus on codes and themes.

Results

From the 33 practice managers contacted, 12 GPs participated in the phone interviews, 11 of whom were female. The GP age range was 33-70 years, all were from the metropolitan area, and the interview time was around 24 min. Two themes were generated from the data: ethical considerations and practical considerations. Ethically, the GP participants discussed barriers including consenting on behalf of a child, parental guilt, and the impact of health insurance, whereas viewing it for health promotion was a facilitator. For practical considerations, barriers included the difficulty in identifying people eligible for screening, maintaining medical communication between service providers, and time and financial constraints, whereas linking screening for telomere length with existing screening and vaccination schedules in South Australia could help increase the uptake of screening while limiting impact on general practice resources.

Ethical considerations

Consenting on behalf of a child

GP participants discussed that there was no guarantee that, should the screening be performed during infancy, that these patients would want to know the results or have consented to the screening later in life. This was contrasted to the childhood immunisation schedule, where the benefits are clearer and less likely to have negative outcomes in adulthood, compared to telomere length screening where the evidence is uncertain.

I guess there's some other ethical things involved in the parents consenting for [screening] but the child might not want them to have known or wanted it out in that sort of public domain, before they were old enough to consent. GP08

The tricky part with this is these are decisions being made by parents on behalf of their children before their children can actually consent. When you're getting to 18/19/20 and you've been presented with this information that you maybe didn't want to know in the first place, that's another element to navigate. GP04

Parental guilt

Participants noted that it may be difficult to manage the potential guilt of parents caused by the results of the screening. As complications in pregnancy (e.g. preeclampsia) can be a result of stigmatised lifestyle factors (e.g. obesity), it may be that a negative outcome from the screening could lead mothers to feel a sense of guilt and responsibility for having potentially contributed poorer health outcomes for their child. Distress to the parent may reduce the net benefit of screening for shortened telomere length in the child.

Guilt is also a concern, as many patients seem to suffer from some kind of guilt during those early years of parenting, this will add to an already existing degree of guilt. GP07

Potentially whether the parent will feel guilt if they do find out ... or [if it] will happen in the future. GP11

I would feel worried that the mum would be made to feel guilty for the pregnancy being a potential cause or risk factor for this child having chronic illness in the future, and often those complications are actually not within the mum's control. GP12

Health insurance

Many participants contemplated what the results of screening for telomere length could have on health insurance within Australia. There was concern that tests identifying shortened telomere length, and therefore higher risks of chronic disease, could be used as justification for health insurance companies to increase costs, or refuse coverage for certain conditions. Participants noted that safeguards would need to be in place to protect the interests of patients.

When you're looking at telomere length and its association with chronic disease maybe, and degenerative diseases later on in life, are the insurers going to grab hold of it – we really don't know. GP01

Insurance may be an issue- will insurance companies insure if a positive test is obtained? GP07

Health promotion

Participants noted that it is important to be able to provide information to people to maximise their ability to make health-altering decisions. This spoke to the role of general practice and primary care, being preventative medicine and monitoring chronic conditions. They noted that some of their patients have an appetite for information that could benefit their health and who would be interested in knowing any risk factors their child may carry. However, this was contrasted with the question of how a patient would be able to action the results of the screening, and the potential connotation that it may relate to genetic screening.

I think the more prevention the better, and the more awareness, the more that we are aware that these things could lead on, it would be fabulous. GP01

It's actually a health promotional tool, I think it's actually very good. However, in the hands of the right people it's actually got a lot of danger. GP06

I think if we can pick up children who are at risk and perhaps, we can intervene...providing the education, via providing referrals through a multi-system, multidisciplinary team...we can treat these children and monitoring them better regularly. GP11

What's the point of screening for something, in some people's minds, if you can't do anything about it? Is there evidence to say that, yes, you can do stuff to reverse it, minimise it? GP03

Practical considerations

Identifying infants eligible for screening

GP participants mentioned that they would find it difficult to identify patients who would be eligible for telomere length screening. Those, in general, practice, unless involved in perinatal care, would be unaware of the conditions of the pregnancy, and therefore less likely to mention it in consults and be more dependent on notes from the obstetrician. This was partly acknowledged as a result of this being an emerging area of research, as well as the difference in skills and interest levels of general practitioners. It was stated that only those with experience in maternal health, and a keen interest in current research would be aware of the impact of telomere length, and the potential to screen for those at risk of negative health outcomes. Furthermore, the indication that there appeared unclear guidelines regarding the follow up of children born after a pregnancy complication was a barrier to identifying and treating a child born after a pregnancy complication.

...hopefully that information would be recorded in our patient record. It would often

be lost, I think ... not necessarily really visible, especially as the child gets older. GP10

Yeah, I think there's always a barrier for the GPs knowing about it, spreading the word to the GPs that don't do the antenatal care, don't keep up to date on these particular things. GP03

No idea if there's guidelines around but any child that's born either premature or a significant complication then basically, I tend to see them at their regular immunisations but also six, nine and then again at 12 months. GP06

Communication between doctors

Participants noted that maintaining streams of communication between treating doctors could be an issue in screening for shortened telomeres. Given that patients may move and/or change between practices, it would be difficult to have the results of the screening follow parents and their children. This barrier was further complicated when considering the timing of the screening. When contemplating screening in infancy, participants noted the custodian of the results to be an important consideration, as the baby is unable to hold the information, and not guaranteed to see the general practitioner who organised the test.

Well, if I had a discharge summary it might be included on that if it's been given to me in a timely manner. I received one last week from six months ago. GP09

We can access My Health Record so we get an idea of what the discharge summary would have said... beyond that it's reliant on history checking. GP05

We'd rely heavily on the obstetrician and on letters. GP08

Time and financial constraints

Almost all participants noted the time limitations within general practice. This was noted in the difference in appointment styles, with some general practitioners valuing longer appointment times, and others providing shorter, bulk-billed appointments. With current limitations, particularly in regional and remote areas, wait lists for non-urgent appointments were noted to be up to 10 weeks.

Absolutely time constraints, that's always an issue in GP Land. That would always be a challenge. GP04

I guess the usual barriers like time management, that's the general one that applies to everywhere not just screening tests, everything we do are always time limited. GP11

Along with time constraints, participants noted that there would need to be financial incentive for screening to be widespread. Some noted that current consultations for antenatal care are poorly remunerated, which is a disincentive for general practitioners to devote more of their time to these issues. Additionally, were the screening to not be a Medicare rebate for doctors, then this cost would be relayed to the parent, which may create equity issues.

Doing those things is time consuming and poorly remunerated but you can say that for all of general practice. GP09 I wonder whether financial costs whether it comes out of the cost to the patient, if it does then, some of them may not want to have it. GP11

Integrating telomere length screening with existing screening

Facilitators included the linking of screening for telomere length to existing appointments in antenatal care. Other appointments where information is provided (such as antenatal appointments) and a health-protective procedure occurs could save time and increase convenience for both general practitioners and parents, increasing the likelihood Downloaded from https://academic.oup.com/fampra/advance-article/doi/10.1093/fampra/cmad064/7188177 by University of Adelaide user on 16 June 2022

of screening. In Australia, the newborn screening test (heel prick to screen for cystic fibrosis, amongst other genetic conditions) and the childhood immunisation schedule (i.e., vaccines given at two, four, and six months), were identified as timepoints in antenatal care within Australia, where telomere length screening could be integrated. The use of a blood test appeared suitable if integrated into current tests, however, the option of a saliva sample would facilitate its use.

When a baby's born, they have the heel prick test for screening of cystic fibrosis and things like that ... Could it be done in hospital before they go as one more test? GP03

These conversations would be happening at the routine immunisation schedule visit ... they could maybe be doing [telomere length screening]. GP02

I guess the easier it is to do the test, like if I can take it in my room with a swab of saliva, great. If I have to send them off to get a blood test, less great. GP09

Nowadays I don't think many people will be opposed to having a blood sample done as long as we explain how we're going to do it, and how we can apply anaesthetic cream for the blood taking. The saliva sample doesn't sound too hard to collect as well. GP11

Discussion

This study explored GPs' perspectives on the potential implementation of telomere length as a neonatal screening test in Australian general practice. Two themes were generated: ethical considerations, including those surrounding parental consent on behalf of the child, parental guilt, the impact to health insurance, but also the impact for health promotion. Practical considerations encompassed the identification of eligible patients, communication, time, and financial constraints, and the integration of screening into current care.

To our knowledge, no other studies have considered telomere length as a screening tool for future chronic diseases in infants, nor have they qualitatively explored the physician's attitudes regarding its potential use. However, several studies in adults have frequently measured telomere length in blood, demonstrating shorter telomeres are associated with increased risk of cardiometabolic disorders.^{13,14} In children, there have been less studies investigating telomere length; however, those that have, consistently show that shorter telomeres are associated with adverse events early in childhood.^{25,26} If a test that measures telomere length in newborns could identify offspring at increased risk of future chronic disease, before disease is apparent in adults, this would have a critical impact on breaking the cycle of future chronic disease risk. Our findings add significant novelty to current literature that has suggested a proposed use, by exploring attitudes and barriers towards its potential uptake as a screening tool.

Critical barriers towards telomere length screening were also raised, with consenting on behalf of the child an important ethical consideration. GPs in the current study were reflective of when offspring at an older age may not have approved of their parents consenting to the newborn screening on their behalf. Since the 1960's, healthcare providers in Australia have offered newborn bloodspot screening to babies in all states and territories. Consent is voluntary and is grounded in a range of ethical concepts, as highlighted in the National Policy Framework, and describes recommendations to protect the privacy of the individual from whom the bloodspot was taken.²⁷ Universally, newborn screening is recognised as one of the most efficient and effective screening programmes.²⁸ However, with increasing opportunities to screen for disorders that previously were difficult to identify in the newborn, this has inherently raised concerns, including those of consent. Specific to genetic testing, consent is based on the benefits for testing outweighing harm and is undoubtedly acceptable. As an example, for Duchenne Muscular Dystrophy, the availability of genomic data requires families to make decisions about information that may predict future events, with differing levels of certainly and the ability to intervene.²⁹ Whereas a disease such as diabetes is attributed to both genetics and lifestyle, individual information about personal risk of developing diabetes in the future may not be appropriate for newborn screening.²⁸ The concerns of GPs are valid, although telomere length is a biomarker of future disease risk and is not a genetic test. The difficulty is balancing the challenges and benefits of screening. This may involve tailored education to GPs on the role of telomere length as a biomarker and the benefits of early screening, but also education and counselling for parents during the pre- and postnatal period.

A novel, though less surprising aspect to this study was the issue of parental guilt. It was raised by the GPs that many parents already feel guilty in the early years of parenting, and so results of such a screening test may pose an additional burden. Moreover, because pregnancy complications may be in-part, attributable to lifestyle factors such as suboptimal maternal nutrition and physical activity, the potential indication of an increased risk for cardiometabolic diseases in later life would further add to parental guilt. The stigma surrounding such a screening test because of potential parental behaviour is unique to our study, though stigma in reproductive health is an emerging concept in the literature.^{30,31} The developmental origins of health and disease, whereby exposures during early life, including *in utero*, influence the risk of later conditions, creates a potential caveat for where individual responsibility lies. Specifically, because maternal behaviours play a significant role in offspring health outcomes, there is potential risk for ongoing maternal blame for adverse offspring health. Most research to date has focussed on weight stigma across pregnancy and antenatal care, revealing associations with psychological distress, decreased access to and uptake of reproductive healthcare, poorer health behaviours, and adverse pregnancy outcomes.³¹ A recent scoping review on 44 studies on stigma related to gestational diabetes, described the feelings of responsibility and guilt were typically related to concerns for the unborn child, and feeling like they failed the unborn child.³² Supportive communication between the doctor-patient and continuity of care may facilitate discussions around emotional experiences and guilt.

Study findings contribute new knowledge by GPs expressing difficulty in identifying infants who might be eligible for screening, specifically those born following a pregnancy complication. It was described that not all GPs are equipped with the knowledge or speciality in maternal health to actively engage with parents/families about the consequences of a complicated pregnancy on the long-term health of off-spring. Such knowledge deficit is not an uncommon theme, as GPs see themselves as generalists and are therefore reluctant to initiate testing considered more specialist.³³ Roth et al.³⁴

identified gaps in healthcare provider knowledge regarding maternal cardiovascular risk after hypertensive disorders during pregnancy, with obstetricians tending to have higher rates of understanding of the risks, but still overall knowledge was lacking.³⁴ Other studies internationally also reveal knowledge gaps among general practitioners in the assessment, management, or diagnosis of women who have perinatal depression³⁵ or vulnerability in pregnancy,³⁶ but even a lack of knowledge in terms of child allergies³⁷ and child obesity.³⁸ Our findings add to the complexity of follow-up screening in both women and their children, where currently, no international consensus for post-partum follow-up care following several pregnancy complications exist,³⁹ or where they do, post-partum follow up is low.⁴⁰ These consistencies highlight the ongoing barriers that GPs regularly face, but strengthen the need for education programs, particularly that are compatible and that have reach, for GPs working in pregnancy and postnatal care, and the relevance of follow-up screening in children.

Strengths and limitations

This study provides an in-depth analysis of the perspectives of GPs to potentially implement a screening test to identify offspring at high risk of future chronic disease. Study findings contribute to the limited knowledge assessing follow-up of screening after pregnancy within Australia and provide novel findings on a potential new screening tool that could be considered early in life. Strengths of this study include the diverse age range and years of GP training, and while all GPs were located within the Adelaide metropolitan area, some of the GPs also had rural experience, enhancing the generalisability of findings. Further research from an international setting would contribute to our findings, along with the potential screening in offspring of culturally and linguistically diverse backgrounds where knowledge and communication gaps are likely larger. There is the possibility that our findings were influenced by desirability bias, where highly motivated and interested GPs were more likely to participate. The views of other health professionals and those of women who had a pregnancy complication may add value to our findings. Finally, despite a relatively small sample size, it was sufficient for data saturation with clear themes generated.

Conclusions

GPs were generally supportive of potential telomere screening in infants, particularly if aligned with current care. However, critical barriers such as consent, the identification of eligible infants, and the potential guilt that women may face, need to be explored and controlled. Our results highlight the need for further exploration of the potential appropriateness of telomere screening among other health professionals, but also parents, and to further investigate how best such a screening test could be implemented into routine care. It also reiterates the need for transparent follow-up guidelines.

Acknowledgements

We are grateful to the interviewed GPs and the GP managers whose participation and support made this study possible.

Ethical Approval

Ethical approval was granted by The University of Adelaide Human Research Ethics Committee [H-2021-110, date approved 26/10/21].

Funding

National Health and Medical Research Council (NHMRC) Ideas Grant was awarded to JAG [APP2000905] and a Women's Health Research Translation Network awarded to JAG.

Conflict of Interest

The authors declare that they have no competing interests.

Data Availability

The data underlying this article cannot be shared publicly due to the privacy of individuals that participated in the study.

Author Contributions

JAG: Funding acquisition, Conceptualisation, Project administration, Writing - Original draft preparation, Writing - Reviewing and Editing. CP: Data curation, Investigation, Writing - Original draft preparation, Writing- Reviewing and Editing. TBM: Conceptualisation, Project administration, Writing- Reviewing and Editing; JM: Methodology, Formal analysis, Writing - Reviewing and Editing.

References

- 1. AIHW. *Health of mothers and babies*. Australian Institute of Health and Welfare; 2022[accessed 2022 Sep 10]. https://www.aihw.gov. au/reports/mothers-babies/health-of-mothers-and-babies.
- Pittara T, Vyrides A, Lamnisos D, Giannakou K. Pre-eclampsia and long-term health outcomes for mother and infant: an umbrella review. *BJOG*. 2021;128(9):1421–1430. doi:10.1111/1471-0528.16683.
- Vounzoulaki E, Khunti K, Abner SC, Tan BK, Davies MJ, Gillies CL. Progression to type 2 diabetes in women with a known history of gestational diabetes: systematic review and meta-analysis. *BMJ*. 2020;369(8245):m1361. doi:10.1136/bmj.m1361.
- Metzger BE, Lowe LP, Dyer AR, Trimble ER, Chaovarindr U, Coustan DR, Hadden DR, McCance DR, Hod M, McIntyre HD, et al.; HAPO Study Cooperative Research Group. Hyperglycemia and adverse pregnancy outcomes. N Engl J Med. May 8 2008;358(19):1991–2002. doi:10.1056/NEJMoa0707943.
- Yazici A, Buyuktiryaki M, Sari FN, Akin MS, Ertekin O, Alyamac Dizdar E. Comparison of different growth curves in the assessment of extrauterine growth restriction in very low birth weight preterm infants. *Arch Pediatr.* Nov 30 2023;30(1):31–35. doi:10.1016/j. arcped.2022.11.008.
- Mol BWJ, Roberts CT, Thangaratinam S, Magee LA, de Groot CJM, Hofmeyr GJ. Pre-eclampsia. *Lancet*. 2016;387(10022):999– 1011. doi:10.1016/S0140-6736(15)00070-7.
- Salam RA, Das JK, Bhutta ZA. Impact of intrauterine growth restriction on long-term health. *Curr Opin Clin Nutr Metab Care*. 2014;17(3):249–254. doi:10.1097/MCO.000000000000051.
- Wicklow BA, Sellers EAC, Sharma AK, Kroeker K, Nickel NC, Philips-Beck W, Shen GX. Association of gestational diabetes and type 2 diabetes exposure in utero with the development of type 2 diabetes in first nations and non-first nations offspring. *JAMA Pediatr.* 2018;172(8):724–731. doi:10.1001/jamapediatrics.2018.1201.

- Turbeville HR, Sasser JM. Preeclampsia beyond pregnancy: longterm consequences for mother and child. *Am J Physiol-Renal Physiol*. 2020;318(6):F1315–F1326. doi:10.1152/ajprenal.00071.2020.
- Mohseni R, Mohammed SH, Safabakhsh M, Mohseni F, Monfared ZS, Seyyedi J, Mejareh ZN, Alizadeh S. Birth weight and risk of cardiovascular disease incidence in adulthood: a dose-response metaanalysis. *Curr Atheroscler Rep.* 2020;22(3):12–12. doi:10.1007/ s11883-020-0829-z.
- Belbasis L, Savvidou MD, Kanu C, Evangelou E, Ioanna T. Birth weight in relation to health and disease in later life: an umbrella review of systematic reviews and meta-analyses. *BMC Med*. 2016;14(1):147–147. doi:10.1186/s12916-016-0692-5.
- D'Mello MJJ, Ross SA, Briel M, Anand SS, Gerstein H, Paré G. Association between shortened leukocyte telomere length and cardiometabolic outcomes. *Circ Cardiovasc Genet*. 2015;8(1):82– 90. doi:10.1161/CIRCGENETICS.113.000485.
- D'Mello MJ, Ross SA, Briel M, Anand SS, Gerstein H, Pare G. Association between shortened leukocyte telomere length and cardiometabolic outcomes: systematic review and metaanalysis. *Circ Cardiovasc Genet*. 2015;8(1):82–90. doi:10.1161/ CIRCGENETICS.113.000485.
- 14. Emami M, Agbaedeng TA, Thomas G, Middeldorp ME, Thiyagarajah A, Wong CX, Elliott AD, Gallagher C, Hendriks JML, Lau DH, et al. Accelerated biological aging secondary to cardiometabolic risk factors is a predictor of cardiovascular mortality: a systematic review and meta-analysis. *Can J Cardiol.* Mar 2022;38(3):365–375. doi:10.1016/j.cjca.2021.10.012.
- 15. Petrou S. The economic consequences of preterm birth during the first 10 years of life. *BJOG*. 2005;112(S1):10–15.
- 16. Farladansky-Gershnabel S, Gal H, Kidron D, Krizhanovsky V, Amiel A, Sukenik-Halevy R, Biron-Shental T. Telomere homeostasis and senescence markers are differently expressed in placentas from pregnancies with early- versus late-onset preeclampsia. *Reprod Sci.* Sep 2019;26(9):1203–1209. doi:10.1177/1933719118811644.
- Toutain J, Prochazkova-Carlotti M, Cappellen D, Jarne A, Chevret E, Ferrer J, Idrissi Y, Pelluard F, Carles D, Maugey-Laulon B, et al. Reduced placental telomere length during pregnancies complicated by intrauterine growth restriction. *PLoS One*. 2013;8(1):e54013e54013. doi:10.1371/journal.pone.0054013.
- McAninch D, Bianco-Miotto T, Gatford KL, Leemaqz SY, Andraweera PH, Garrett A, Plummer MD, Dekker GA, Roberts CT, Smithers LG, et al. The metabolic syndrome in pregnancy and its association with child telomere length. *Diabetologia*. 2020;63(10):2140–2149. doi:10.1007/s00125-020-05242-0.
- Otlowski M, Tiller J, Barlow-Stewart K, Lacaze P. Genetic testing and insurance in Australia. Aust J Gen Pract. 2019;48(3):96–99.
- 20. Garg P, Ha MT, Eastwood J, Harvey S, Woolfenden S, Murphy E, Dissanayake C, Williams K, Jalaludin B, McKenzie A, et al. Health professional perceptions regarding screening tools for developmental surveillance for children in a multicultural part of Sydney, Australia. BMC Fam Pract. 2018;19(1):42. doi:10.1186/s12875-018-0728-3.
- Carlton J, Griffiths HJ, Horwood AM, Mazzone PP, Walker R, Simonsz HJ. Acceptability of childhood screening: a systematic narrative review. *Public Health*. 2021;193(4):126–138. doi:10.1016/j. puhe.2021.02.005.
- 22. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *Int J Qual Health Care*. 2007;19(6):349–357. doi:10.1093/intqhc/mzm042.
- Barrett AKA, Johnston J. How to ... be reflexive when conducting qualitative research. *Clin Teacher*. 2020;17(1):9–12. doi:10.1111/ tct.13133.
- 24. Braun V, Clarke V. Using thematic analysis in psychology. *Qual Res Psychol*. 2006;3(2):77–101. doi:10.1191/1478088706qp063oa.
- 25. Lang J, McKie J, Smith H, McLaughlin A, Gillberg C, Shiels PG, Minnis H. Adverse childhood experiences, epigenetics and telomere length variation in childhood and beyond: a systematic review of

the literature. Eur Child Adolesc Psychiatry. 2020;29(10):1329–1338. doi:10.1007/s00787-019-01329-1.

- Soares S, Rocha V, Kelly-Irving M, Stringhini S, Fraga S. Adverse childhood events and health biomarkers: a systematic review. *Front Public Health*. 2021;9:649825. doi:10.3389/fpubh.2021.649825.
- Department of Health. Newborn bloodspot screening. National Policy Framework. Australian Health Ministers' Advisory Council; 2018 [accessed 2022 Sep 10]. https://www.health.gov.au/sites/ default/files/documents/2020/10/newborn-bloodspot-screeningnational-policy-framework.pdf.
- Kerruish NJ, Robertson SP. Newborn screening: new developments, new dilemmas. J Med Ethics. 2005;31(7):393–398. doi:10.1136/ jme.2004.008219.
- 29. Landrum Peay H, Fischer R, Tzeng JP, Hesterlee SE, Morris C, Strong Martin A, Rensch C, Smith E, Ricotti V, Beaverson K, et al. Gene therapy as a potential therapeutic option for Duchenne muscular dystrophy: A qualitative preference study of patients and parents. *PLoS ONE*. 2019;14(5):e0213649. doi:10.1371/journal. pone.0213649.
- 30. Bohren MA, Vazquez Corona M, Odiase OJ, et al. Strategies to reduce stigma and discrimination in sexual and reproductive healthcare settings: a mixed-methods systematic review. *PLOS Glob Public Health*. 2022;2(6):e0000582. doi:10.1371/journal. pgph.0000582.
- Hill B, Incollingo Rodriguez AC. Weight stigma across the preconception, pregnancy, and postpartum periods: a narrative review and conceptual model. *Semin Reprod Med.* 2020;38(6):414–422. doi:10.1055/s-0041-1723775.
- 32. Davidsen E, Maindal HT, Rod MH, Olesen K, Byrne M, Damm P, Nielsen KK. The stigma associated with gestational diabetes mellitus: a scoping review. *EClinicalMedicine*. 2022;52:101614. doi:10.1016/j.eclinm.2022.101614.
- 33. Standing HC, Jarvis H, Orr J, Exley C, Hudson M, Kaner E, Hanratty B. GPs' experiences and perceptions of early detection of liver disease: a qualitative study in primary care. Br J Gen Pract. 2018;68(676):e743-e749. doi:10.3399/bjgp18X699377.
- 34. Roth H, LeMarquand G, Henry A, Homer C. Assessing knowledge gaps of women and healthcare providers concerning cardiovascular risk after hypertensive disorders of pregnancy—a scoping review. systematic review. *Front Cardiovasc Med.* 2019;6:178. doi:10.3389/fcvm.2019.00178.
- 35. Ford E, Shakespeare J, Elias F, Ayers S. Recognition and management of perinatal depression and anxiety by general practitioners: a systematic review. *Fam Pract.* 2017;34(1):11–19. doi:10.1093/ fampra/cmw101.
- 36. Brygger Veno L, Pedersen LB, Sondergaard J, Ertmann RK, Jarbol DE. Assessing and addressing vulnerability in pregnancy: general practitioners perceived barriers and facilitators a qualitative interview study. *BMC Prim Care*. 2022;23(1):142. doi:10.1186/s12875-022-01708-9.
- 37. Madrazo JA, Alrefaee F, Chakrabarty A, de Leon JC, Geng L, Gong S, Heine RG, Järvi A, Ngamphaiboon J, Ong C, et al. International cross-sectional survey among healthcare professionals on the management of cow's milk protein allergy and lactose intolerance in infants and children. *Pediatr Gastroenterol Hepatol Nutr.* 2022;25(3):263–275. doi:10.5223/pghn.2022.25.3.263.
- 38. Hennessy M, Byrne M, Laws R, Mc Sharry J, O'Malley G, Heary C. Childhood obesity prevention: priority areas for future research and barriers and facilitators to knowledge translation, coproduced using the nominal group technique. *Transl Behav Med*. 2019;9(4):759–767. doi:10.1093/tbm/iby074.
- 39. Grieger JA, Hutchesson MJ, Cooray SD, et al. A review of maternal overweight and obesity and its impact on cardiometabolic outcomes during pregnancy and postpartum. *Ther Adv Reprod Health*. 2021;15:eCollection. doi:10.1177/2633494120986544.
- Martinez NG, Niznik CM, Yee LM. Optimizing postpartum care for the patient with gestational diabetes mellitus. Am J Obstet Gynecol. 2017;217(3):314–321. doi:10.1016/j.ajog.2017.04.033.